Author’s response to reviews

Title: Microsphere-based antibody assays for human parvovirus B19V, CMV and T. gondii

Authors:

Yilin Wang (yilin.wang@helsinki.fi)
Lea Hedman (lea.hedman@helsinki.fi)
Maria F. Perdomo (maria.perdomo@helsinki.fi)
Amal Elfaitouri (amalelfaitouri@hotmail.com)
Agnes Bölin-Wiener (agneswiener@gmail.com)
Arun Kumar (arun.q.kumar@gsk.com)
Maija Lappalainen (Maija.Lappalainen@hus.fi)
Maria Söderlund-Venermo (Maria.Soderlund-Venermo@helsinki.fi)
Jonas Blomberg (jonas.blomberg@medsci.uu.se)
Klaus Hedman (Klaus.Hedman@helsinki.fi)

Version: 2 Date: 28 Sep 2015

Author’s response to reviews:

Helsinki, September 25, 2015

Dear Danielle Talbot:

We thank you warmly for your positive attitude towards our manuscript. Below is our response (in red font) to your request (in black). Enclosed is the re-revised manuscript with the latest revisions highlighted yellow. In addition to the one requested (below), we did two updatings of address/funding (lines 11-12 and 303).

Yours sincerely,

Yilin Wang
University of Helsinki, Department of Virology
Editorial Requests:

1. Clarify in the Methods if all samples were taken as part of standard care or if the authors obtained consent from the patients

This is now clarified in Methods. Please see lines 175-180.